Author's response to reviews

Title: Diagnostic criteria for diabetes revisited: Making use of combined criteria

Authors:

Ali Parappil (aliparappil@yahoo.com)
Suhail A.R. Doi (sardoi@onebox.com)
Kamal A.S. Al-Shoumer (kshoumer@hsc.kuniv.edu.kw)

Version: 8 Date: 17 Jan 2002

Dear Sir

We have realized that Dr Schriger is right about the information from bar graphs and thus we have decided to implement boxplots rather than bar graphs as originally suggested by Dr Schriger. Figures 1a/b and 2a/b are now replaced by figures 1-4 box plots. The legend has been updated. Everything else remains the same.

Kind regards

Dr Suhail Doi